Author's response to reviews

Title: Recovery of NIS expression in thyroid cancer cells by overexpression of Pax8 gene

Authors:

Ivan Presta (presta@unicz.it)
Franco Arturi (arturi@unicz.it)
Elisabetta Ferretti (elferrettiti@tin.it)
Tiziana Mattei (tiziana.mattei@uniroma1.it)
Daniela Scarpelli (dascar4@yahoo.it)
Emanuele Tosi (emanuele.tosi@uniroma1.it)
Angela Scipioni (angela.scipioni@tiscali.it)
Marilena Celano (celano.donato@tiscali.it)
Alberto Gulino (alberto.gulino@uniroma1.it)
Sebastiano Filetti (sebastiano.filetti@uniroma1.it)
Diego Russo (d.russo@unicz.it)

Version: 3 Date: 12 July 2005

Author's response to reviews: see over
To the Editor:

We would like to submit the revised version of manuscript # MS: 9786323936682327 (Recovery of NIS expression in thyroid cancer cells by overexpression of Pax8 gene) for publication in BMC cancer. The paper has been formatted according to the requested changes.

We have outlined all changes made in the manuscript in the point-by-point description reported below in this letter. The revised manuscript has been submitted electronically according to your guidelines.

CHANGES MADE

Affiliations: revised;

Corresponding author: indicated on the title page;

Author list: revised;

Keywords section: removed;

Page breaks: removed;

Materials & Methods: section renamed;

Competing interests: made;

Figures: revised number 3, 5 and 6;

Text: revised in legend to figures section.

Sincerely yours,

Sebastiano Filetti, MD

Università degli Studi Roma “La Sapienza”
Dipartimento Scienze Cliniche- Clinica Medica 2 Viale del Policlinico, 155
00161 Roma, Italy - Phone: 39-06-49978304 Fax : 39-06-4463783
e-mail: sebastiano.filetti@uniroma1.it